Author's response to reviews

Title: Measuring health-related quality of life in men with osteoporosis or osteoporotic fracture

Authors:

Marta Zwart (mzwart.girona.ics@gencat.cat)
Rafael Azagra (rafael.azagra@uab.cat)
Gloria Encabo (gencabo@vhebron.net)
Amada Aguye (amyaguye@telefonica.net)
Genís Roca (groca@movistar.es)
Sílvia Güell (24965sgp@comb.es)
Núria Puchol (39434npr@comb.cat)
Emili Gene (29833egt@comb.cat)
Francesc López-Expósito (flopeze@papps.org)
Sílvia Solà (ssolamun@yahoo.es)
Sergio Ortiz (sergibdv@hotmail.com)
Pilar Sancho (sanchomp@gmail.com)
Liz Abado (liarcv97@hotmail.com)
Milagros Iglesias (milaiglesias@camfic.org)
Pere Torán (ptoran.bnm.ics@gencat.cat)
Adolf Diez-Perez (ADiez@imas.imim.es)

Version: 6 Date: 28 September 2011

Author's response to reviews:

To BMC Production Team,

Thank you very much for your help.

I have revised the manuscript and changed the points listed below and then uploaded it again.

- Reference 12 in order within the manuscript.

- I have changed in the abstract 'Methods and Design' to 'Methods/Design' and removed the additional 'Design' heading. Also updated the abstract in the submission system.

- Placed Appendix directly before the references.

- Removed Additional Files